Author’s response to reviews

Title: A novel compound mutation in GLRA1 cause hyperekplexia in a Chinese boy- a case report and review of the literature

Authors:
Zhiliang Yang (sizhewujiu@163.com)
Guilian Sun (sungl63@hotmail.com)
Fang Yao (yaofangmail@hotmail.com)
Dongying Tao (1754887753@qq.com)
Binlu Zhu (249308925@qq.com)

Version: 2 Date: 12 Sep 2017

Author’s response to reviews:

2017-9-12

Dear Dr. Matteo Pasini,

We are grateful for the opportunity to revise our paper (MGTC-D-17-00157R1) entitled” A novel compound mutation in GLRA1 cause hyperekplexia in a Chinese boy- a case report and review of the literature”, and the helpful comments of your reviewers.

We attach a version showing the tracked changes in red words and, separately list our point-by-point responses.

Yours sincerely,

Zhiliang Yang,MD,PhD.
Department of Pediatrics,
The First Hospital of China Medical University,
110001,No. 155 Nanjing North Street, Heping District, Shenyang, Liaoning, China.
Tel: +86 24 8328 2527;E-mail: sizhewujiu@163.com
Answer to reviewers:

Technical Comments:

1. Authors' contributions

Please use initials to refer to each author's contribution in this section instead of full names. Also, please add the following sentence at the end of the statement: "All authors read and approved the final manuscript."

Answer: We have edited the part according to the comments.

2. Tables

Please remove the tables from the manuscript and upload them as separate files without the legends.

Answer: We have edited the table according to the comments.

3. Table legends

Please add the table legends to the Figure legends section (page 17) and change the header to 'Table and figure legends'.

Answer: We have edited the part according to the comments.

4. Figure legends

Please remove the legends from the figure files as they are already listed in the Table and figure legends section.

Answer: We have edited the figures according to the comments.

5. Supplementary material

Please remove the CARE checklist from the supplementary material as this file will not be included in the published article.
Answer: We have removed the CARE checklist.

Reviewer reports:

Márta Széll (Reviewer 1): The Authors carefully considered all my recommendations and corrections and built into the revised version of the manuscript. I have no further recommendations or corrections.

Answer: We very appreciated the comments.